Author’s response to reviews

Title: Scheduling medicines as controlled substances: Addressing normative and democratic gaps through human rights-based analysis

Authors:

Diederik Lohman (diederiklohman@gmail.com)

Damon Barrett (damon.barrett@gu.se)

Version: 2 Date: 04 Apr 2020

Author’s response to reviews:

We have added a list of abbreviations used in the manuscript and relevant information on ethics review, conflict of interest, acknowledgements etc.